Author’s response to reviews

Title: Drug utilization patterns and reported health status in ethnic German migrants (Aussiedler) in Germany: a cross-sectional study

Authors:

Anna Volodina (anna.volodina@yahoo.com)
Thilo Bertsche (thilo.bertsche@uni-leipzig.de)
Karel Kostev (kkostev@de.imshealth.com)
Volker Winkler (Volker.Winkler@urz.uni-heidelberg.de)
Walter Emil Haefeli (Walter.Emil.Haefeli@med.uni-heidelberg.de)
Heiko Becher (heiko.becher@urz.uni-heidelberg.de)

Version: 3 Date: 19 May 2011

Author’s response to reviews: see over
Manuscript title:

Drug utilization patterns and reported health status in ethnic German migrants

(Aussiedler) in Germany: a cross-sectional study

Authors:

Anna Volodina, Thilo Bertsche, Karel Kostev, Volker Winkler, Walter Emil Haefeli, Heiko Becher

Response to reviewers II

19 May 2011
Reviewer: Maninder Singh Setia

Reviewer’s report:  
The authors have satisfactorily responded to the earlier comments. Barring some minor typos (editing) the manuscript should be fine.

Level of interest: an article whose findings are important to those with closely related research interests

Quality of written English: Needs some language corrections before being published

Answer: We agree, typos have been corrected and minor editing of the manuscript was done